Author’s response to reviews

Title: Post-operative re-bleeding in patients with hypertensive ICH is closely associated with the CT blend sign

Authors:

Guofeng Wu (wuguofeng3013@sina.com)
Zhengkui Shen (420339169@qq.com)
Likun Wang (769070308@qq.com)
Shujie Sun (sunshujie11@126.com)
Jinbiao Luo (ljbdoctor@126.com)
Yuanhong Mao (961766707@qq.com)

Version: 2 Date: 11 Jun 2017

Author’s response to reviews:

Dear editor

The manuscript has been edited again by the American Journal of Experts. The English level should have been elevated. The revised parts in the text were marked red.

Editor Comments:

BMC Neurology operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Responses-We have viewed the reports and revised the manuscript in the light of the first reviewer’s comments. The second reviewer has no comments for NURL-D-17-00033R1
Reviewer reports:

Marc Alain Babi, MD (Reviewer 1): The authors addressed most of the concerns that I have previously raised.

However, I would still recommend working of the verbiage as they are still numerous sentences that are hard to follow or do no clearly connect to each other.

Otherwise, interesting study, and will further add to the body of literature.

Responses-The manuscript (NURL-D-17-00033R1) has been edited again by the American Journal Experts. The English level must have been improved greatly. The editorial certificate of the manuscript was also provided.